Author’s response to reviews

Title: Ureteral tumor in an ectopic duplex system: A case report

Authors:

Omar Karray (omar.karray.88@gmail.com)
Hassen Khouni (hassenelkhouni@gmail.com)
Mahdi Charfi (mahdi.charfi.79@gmail.com)
Rami Boulma (rboulma@hotmail.fr)
Mehdi Debaibi (meh_debaibi_90@yahoo.fr)
Rym Makhlouf (rym.makhlouf.91@gmail.com)
Karim Bargaoui (karim.bargaoui@gmail.com)
Oumeima Nessej (oumeima.mesaadinessej@hotmail.fr)
Azza Seridi (azza.seridi@gmail.com)
Slim Fourti (slim.fourti.fsi@rns.tn)
Mohamed Habib Bouhaouala (Mhabib.bouhaouala@rns.tn)
Adnene Chouchene (adnene.chouchene@yahoo.fr)

Version: 1 Date: 16 Dec 2018

Author’s response to reviews:

A revision of the english language level, with a correction of medical terms has been performed by the english language section of the Faculty of Medecine of Tunis.

Medical terminology and figures legends has been revised.
The description of radiological anomalies has been rectified.
Figures legend has been revised and corrected.